Author's response to reviews

Title: Laterally wedged insoles in knee osteoarthritis: do biomechanical effects decline after one month of wear?

Authors:

Rana S Hinman (ranash@unimelb.edu.au)
Kelly Bowles (kbowles@unimelb.edu.au)
Kim L Bennell (k.bennell@unimelb.edu.au)

Version: 3 Date: 9 November 2009

Author's response to reviews:

Dear Editors,

The following changes have been made to the manuscript as instructed:

1. A conclusions section has been added after the discussion

2. With the exception of two papers that have more than 10 authors, all authors names have been provided in the reference list

3. Authors qualifications have been removed from the manuscript

4. The funding statement has been removed from the title page and has been added to the acknowledgement section

5. The conflict of interest statement has been removed from the title page

6. The running head has been removed

7. Email addresses have been included for all authors on the title page

8. The highlighting has been removed from the manuscript

9. The keywords have been removed

10. The table and each of the three figures have been uploaded as separate files and have been deleted from the manuscript. Legends and titles remain listed in the manuscript after the references.